Author's response to reviews

Title: Resveratrol protects leukemic cells against cytotoxicity induced by proteasome inhibitors via induction of FOXO1 and p27Kip1

Authors:

Xiao-Fang Niu (niuxiaofang521@126.com)
Bao-Qin Liu (baoqinliu@163.com)
Zhen-Xian Du (dzx_doctor@hotmail.com)
Yan-Yan Gao (angela_042214@163.com)
Chao Li (lichao@hotmail.com)
Ning Li (lining@hotmail.com)
Yifu Guan (yfguan@mail.cmu.edu.cn)
Hua-Qin Wang (wanghq_doctor@hotmail.com)

Version: 7 Date: 9 March 2011

Author's response to reviews:

The "Competing interests" section:
The previous version was "The authors declare that have no", "competing interests" was missed.
The correct one should be "The authors declare that have no competing interests."